

**Author's response to reviews**

**Title:** Pilot Evaluation of the text4baby mobile health program

**Authors:**

William D Evans ([wdevans@gwu.edu](mailto:wdevans@gwu.edu))  
Jasmine L Wallace ([jwallace@gwu.edu](mailto:jwallace@gwu.edu))  
Jeremy Snider ([jeremy.snider@gmail.com](mailto:jeremy.snider@gmail.com))

**Version:** 5 **Date:** 27 July 2012

**Author's response to reviews:**

The revised manuscript is in response to the editorial office's email to the corresponding author dated 27 July 2012. Thank you.